vc/26-Aug-03
Week 35 starting 25 Aug 03

SM18 Operation SCHEDULE Meeting 

Schedule for this week : 
	BENCH
	CURRENT

MAGNET
	STATUS 

 Today
	Forward look,   & Next  Magnet  

	TBA1
	2020
	cold
	Disconnect Wed 27th ?

Bring in 2028

	TBA2
	2017
	warm
	 Rapid connection & Tests 

	TBC1
	3041
	warm
	Disconnection & …..  Cluster blocked by SSS but still bring in 1031 

	TBC2
	SSS1
	Warm 
	Mech. Mods. To prepare for SSS1 then, Connection of SSS1

	TBE1
	2018
	cold
	+cold tests + Str.Wire +warm up by 27th

	TBE2
	
	Warm
	1019 removal & arrival of 3046 as soon as possible

	TBF1
	2025 proposed 
	
	Prepare For 

commissioning

	TBF2
	2025 after F1 commissioned
	
	Prepare For 

commissioning


· Test Coordinator this week(35):   Marco

· MTA Technician on call:  (16-3602) Giorgio

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
· Problems from Operation Last week (Stephane + Test Supervisor )
	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Monday 18_M
	AQA alert MITTS too high
	MTM
	
	OK from test coordinator.

	Monday 18_A
	MMP program not responding. Software blocked.
	CO
	MMP
	TEMA restarted.

	Monday 18_N
	IAP@cold
	CO
	TEMA
	Not Resolved

	Monday 18_N
	TBA2:Cryo OK loss during loadline
	MTM
	
	Restart the load line.

	Monday 18_N
	TBA2 : RRR+6.5 kA min energy

Sun blocked. AFS server authentification fault.Reboot impossible.

LF files lost


	MTM
	LF
	Action :M.Bjork 

	Monday 18_N
	TBC1 : Quench @1.5 kA : Sun MTA20 down

AFS server authentification fault
	CO
	
	Action :M.Bjork

	Monday 18_N
	RRR: time in file does not correspond to the time of quench (3 hours of difference)
	
	
	

	Tuesday 19_M
	AQA alert MIITS too high (6.5 kA min energy)
	MTM
	
	OK from test coordinator.

	Tuesday 19_M
	TBC1 : quench at 1.5 kA .IDDCT data not present 
	CO
	AQA
	Data present but Problem with AQA.

	Tuesday 19_M
	Use of DVN_DAQ 2 during RRR can induce a crash. 
	CO
	
	Do not use DVM DAQ 2 but DVM_DAQ

	Tuesday 19_M
	DVM DAQ not working. 
	CO
	
	Repaired by Adrian.

	Tuesday 19_N
	Joint measurements: Transfer the data from the local sun. 
	MTM
	
	

	Wednesday 20_A
	Cryo OK lost during load line.

BP too high 1500 mbar : Liquefactor cannot reduce pressure (BA7 was working)
	ACR
	
	 ACR informed the liquefactor that the pressure is too high.

	Wednesday 20_A
	Magnet change in TBC1 but application HF , LF do not follow. 
	MTM/CO
	HF,LF
	 Do we have to close the application before?

	Wednesday 20_N
	Joints measurement (TBC1) : Explain how to check the data and why it was not correctly done (Claude, Ram, Daniel) 
	MTM
	DVM DAQ
	 P.Pugnat will explain again?

	Wednesday 20_N
	Main current leads disconnection: done by OP team because meca.team is not responding
	MTM
	
	 Decision to be anticipated by the test coordinator?

	Wednesday 20_N
	Quench at 4.2 K: Quench in V clamp and MIITS close to upper limit. 
	MTM/CO
	AQA
	 Test coordinator decision

	Thursday 21_A
	RRR (TBC1) : file not correct
	CO
	AQA
	 Solved by Adrian.

	Thursday 21_N
	Quench at 1.5 kA (all LF) only 2 quenches discharges.
	MTM
	
	One card not triggered?

Y removed for these tests.

	Friday 22_A
	HF DAQ not armed during quench at 3 kA (min energy): AQA said faulty quench.
	MTM
	
	

	Friday 22_N
	Tcoil problem (TBA1): short circuit or faulty coil 
	MTM
	
	Changing the power supply?

	Saturday 23_A
	MM : shaft position (TBA1): rack still in PDI configuration.
	MTM
	
	OK by equip.support











